Author's response to reviews

Title: Antilock Breaking System Effectiveness in Prevention of Road Traffic Crashes in Iran

Authors:

Davoud Khorasani Zavareh (davoud.khorasani@gmail.com)
Saeed Shoar (ssht84@yahoo.com)
Soheil Saadat (soheil.saadat@gmail.com)

Version: 4 Date: 6 March 2013

Author's response to reviews:

Dear Editor-in-Chief,

Please find attached a revised version of our article entitled “Antilock Breaking System Effectiveness in Prevention of Road Traffic Crashes in Iran” with MS: 6942368567439124.

Thank you for giving us the opportunity to revise our manuscript again. Moreover, we appreciate reviewer’s close attention on the manuscript and his minor corrections on that.

All minor revisions advised by the reviewer are made and hope that, in the revised form, you will find it acceptable for publication.

Yours sincerely,

Soheil Saadat
Corresponding author